Author's response to reviews

Title: Rare ocular features in a case of Kabuki syndrome (Niikawa-Kuroki syndrome)

Authors:
- Yi-Hsing Chen (yihsing@gmail.com)
- Ming-Hui Sun (minghui0215@gmail.com)
- Shao-Hsuan Hsia (shsia@adm.cgmh.org.tw)
- Chi-Chun Lai (chichun.lai@gmail.com)
- Wei-Chi Wu (weichi666@gmail.com)

Version: 4
Date: 11 January 2014

Author's response to reviews: see over
Author's covering letter for initial submission

Title: Rare ocular features in a case of Kabuki syndrome (Niikawa-Kuroki syndrome)

Authors:

Version: 1 Date: 20 May 2013

Comments: see over
Dear Dr. Aime:

Enclosed is a manuscript entitled “Rare ocular features in a case of Kabuki syndrome (Niikawa-Kuroki syndrome)”, which we would like you to consider for publication in your journal, BMC Ophthalmology (Section of Case Reports).

Kabuki syndrome is a rare multi-system syndrome with peculiar facial features. Ocular abnormalities that may affect vision are often present in patients with Kabuki syndrome. To our knowledge, our case represents the first report of a patient of Kabuki syndrome with the rare finding of a dysplastic and elevated disc without central cupping. Detailed ophthalmic evaluation of all children with Kabuki syndrome is advised.

As corresponding author, I confirmed by this statement that I had full access to all the data in this report and take responsibility for the decision to submission. There has been no duplicate submission in other journal. We believe that the information from this study is helpful for clinicians and should be considered for publication as a case report in *BMC Ophthalmology*.
Sincerely,

Wei-Chi Wu, MD, PhD

Department of Ophthalmology, Chang Gung Memorial Hospital.

No. 5, Fu-Hsing Street, Kweishan, Taoyuan, 333, Taiwan

TEL: 886-3-3281200, ext. 8666

FAX: 886-3-3287798

E-mail: weichi666@gmail.com